Author's response to reviews

Title: Exploring Diabetes Burden in Mongolia Through a National Knowledge, Attitudes and Practices Survey

Authors:

Alessandro R Demaio (demaio@sund.ku.dk)
Dugee Otgontuya (otgon_phi@hotmail.com)
Maximilian de Courten (maxc@sund.ku.dk)
Ib C Bygbjerg (iby@sund.ku.dk)
Palam Enkhtuya (p_enkhee2001@yahoo.com)
Janchiv Oyunbileg (jobileg@gmail.com)
Dan W Meyrowitsch (dame@sund.ku.dk)

Version: 4 Date: 19 February 2013

Author's response to reviews: see over
Dear Reviewers,

Thank you for your continued support and feedback.

Reviewer 1

Major Compulsory Revisions

1. The first paragraph of the discussion section should really highlight the results of this study, not those of another study.
   Thank you. This has been amended.

2. In the Methods section on page the authors refer to regression statistics and separately to multivariate regression in two different paragraphs, but this could all probably be combined into one single paragraph and description for clarity.
   This has been combined as requested.

Minor Essential Revisions

1. Table 1 now shows column percent, but the column labels indicate that the percentages are the percent of total, please fix.
   Thank you, this has been revised.

2. There are still some minor grammatical errors that need to be fixed
   An English language specialist has reviewed the paper again.

Reviewer 2

The authors have addressed all reviewer comments and have edited the manuscript as requested. I am of the opinion that this manuscript is now ready for publication and should be of interest to those working within this area of care.
Thank you.